Published in final edited form as:

Int J Eat Disord. 2022 October; 55(10): 1397–1403. doi:10.1002/eat.23777.

Differential comorbidity profiles in avoidant/restrictive food intake disorder and anorexia nervosa: Does age play a role?

P. Evelyna Kambanis, MS¹, Stephanie G. Harshman, PhD^{2,4}, Megan C. Kuhnle, BA^{2,4}, Danielle L. Kahn, BA^{2,4}, Melissa J. Dreier, BA^{1,2}, Kristine Hauser, MSN⁴, Meghan Slattery, MSN⁴, Kendra R. Becker, PhD^{2,3}, Lauren Breithaupt, PhD^{2,3}, Madhusmita Misra, MD, MPH^{4,10,11}, Nadia Micali, MD, PhD^{6,7,8,9}, Elizabeth A. Lawson, MD, MMSc^{4,5,*}, Kamryn T. Eddy, PhD^{2,3,*}, Jennifer J. Thomas, PhD^{2,3,*}

¹Department of Psychology, University of Wyoming, Laramie, WY

²Eating Disorders Clinical and Research Program, Massachusetts General Hospital, Boston, MA

³Department of Psychiatry, Harvard Medical School, Boston, MA

⁴Neuroendocrine Unit, Department of Medicine, Massachusetts General Hospital, Boston, MA

⁵Department of Medicine, Harvard Medical School, Boston, MA

⁶Department of Psychiatry, University of Geneva, Switzerland

⁷Department of Pediatrics, University of Geneva, Switzerland

⁸Department of Gynecology and Obstetrics, University of Geneva, Switzerland

⁹GOSH Institute, University College London, UK

¹⁰Division of Pediatric Endocrinology, Massachusetts General Hospital, Boston, MA

Abstract

Objective: Research comparing psychiatric comorbidities between individuals with avoidant/ restrictive food intake disorder (ARFID) and anorexia nervosa (AN) is limited. ARFID often develops in childhood whereas AN most typically develops in adolescence or young adulthood. Understanding how age may impact differential psychological comorbidity profiles is important to inform etiological conceptualization, differential diagnosis, and treatment planning. We aimed to

Public significance statement: Our results highlight that, with the exception of suicidality, which was three times less common in ARFID than AN irrespective of age, observed differences in psychiatric comorbidities in clinical practice may reflect ARFID's younger age at clinical presentation compared to AN.

Conflicts of interest/disclosures: Drs. Thomas, Becker, and Eddy receive royalties from Cambridge University Press for the sale of their books about avoidant/restrictive food intake disorder. Dr. Misra has consulted for Sanofi and Abbvie and served on the Scientific Advisory Board of Abbvie and Ipsen. Dr. Lawson served on the scientific advisory board and has a financial interest in OXT Therapeutics, a company developing oxytocin-based therapeutics for obesity and metabolic disease. Dr. Lawson's interests were reviewed and are managed by MGH and Mass General Brigham (f/k/a/Partners Healthcare) in accordance with their conflict of interest policies. Dr. Breithaupt is a consultant for HealthiVibe, a division of CorEvitas.

¹¹Department of Pediatrics, Harvard Medical School, Boston, MA

^{*}Corresponding author: Jennifer J. Thomas, Ph.D., Eating Disorders Clinical & Research Program, Massachusetts General Hospital, 2 Longfellow Place, Suite 200, Boston, MA 02114; Phone: (617) 643-6306; jjthomas@mgh.harvard.edu.

compare the lifetime frequency of psychiatric comorbidities and suicidality between participants assigned female at birth with ARFID (n=51) and AN (n=40), investigating the role of age as a covariate.

Method: We used structured interviews to assess the comparative frequency of psychiatric comorbidities/suicidality.

Results: When age was omitted from analyses, females with ARFID had a lower frequency of depressive disorders and suicidality compared to AN. Adjusting for age, only suicidality differed between groups.

Discussion: This is the first study to compare comorbidities using a relatively even split of ARFID and AN, and a structured clinical interview to confer ARFID and comorbidities, covarying for age, and the first to compare suicidality. Although suicidality is at least three times less common in ARFID than AN, observed differences in other psychiatric comorbidities may reflect ARFID's relatively younger age of presentation compared to AN.

Keywords

avoidant/restrictive food intake disorder; ARFID; anorexia nervosa; psychiatric comorbidities; depressive and bipolar-related disorders; anxiety, obsessive-compulsive, and trauma-related disorders; neurodevelopmental, disruptive, and conduct disorders; suicidality; structured clinical interview; feeding and eating disorders

Avoidant/restrictive food intake disorder (ARFID) and anorexia nervosa (AN) are characterized by dietary avoidance/restriction. While avoidance/restriction in ARFID can be motivated by sensory sensitivity, fear of aversive consequences, and/or lack of interest in eating, in AN, it is typically driven by shape/weight concerns (APA, 2013; Thomas & Eddy, 2019). Data comparing psychiatric comorbidities between ARFID and AN are limited. Such research is warranted to inform etiological conceptualization, differential diagnosis, and treatment planning.

Findings on the frequency of comorbidities in ARFID versus AN are mixed. Comorbid depressive disorders appear less common in ARFID (Aulinas et al., 2020; Bryson et al., 2018; Fisher et al., 2014; Keery et al., 2019; Lieberman et al., 2019; Nicely et al., 2014). No studies have compared suicidality. Anxiety and neurodevelopmental disorders appear more common in ARFID (Aulinas et al., 2020; Bryson et al., 2018; Fisher et al., 2014; Keery et al., 2019; Lieberman et al., 2019; Nicely et al., 2014; Ornstein et al., 201; Zanna et al., 2020). However, studies comparing psychiatric comorbidities are limited by unbalanced designs/small sample sizes, reliance on treatment-seeking samples, lack of structured interviews, use of retrospective chart reviews and self-report to confer diagnoses, and absence of diagnostic reliability information.

In the above studies, ARFID samples were younger than AN samples, highlighting the possibility that observed differences in comorbidities may be attributed to age. Indeed, only one study (Aulinas et al., 2020) adjusted for age. This is critical given the well-known ages of onset of specific psychiatric disorders, resulting from brain development/activation of pubertal hormones and age-related changes in psychosocial influences. In a large-scale

meta-analysis of epidemiological studies, Solmi and colleagues (2021) reported on peak ages of onset: neurodevelopmental disorders onset earliest in life (5.5 years), followed by anxiety disorders (5.5 years for specific phobia/separation anxiety disorder, 14.5 years for social anxiety disorder, and 15.5 years for panic/generalized anxiety disorder), and depressive disorders (20.5 years). Given younger age of onset for ARFID than AN (e.g., Becker et al., 2019), understanding how age may impact differential comorbidity profiles is important.

We compared the frequency of lifetime psychiatric comorbidities and suicidality between children/adolescents assigned female at birth with ARFID and AN, investigating the role of age as a covariate. For Aim 1, we omitted age as a covariate to replicate findings from extant research, forming hypotheses based on these findings. We also wanted to examine how findings compared with and without adjusting for age. We hypothesized that females with ARFID would demonstrate: (a) lower frequency of depressive and bipolar-related disorders ("depressive disorders") and suicidality; (b) higher frequency of anxiety, obsessive-compulsive, and trauma-related disorders ("anxiety disorders"); and (c) higher frequency of neurodevelopmental, disruptive, and conduct disorders ("neurodevelopmental disorders"). For Aim 2 (including age as a covariate), we had no a priori hypotheses and conducted exploratory analyses in the spirit of discovery (Kraemer, 2015).

Method

Participants

We recruited participants (N= 91; n = 51 ARFID; n = 40 AN), ages 10–23 years, for two studies of the neurobiology of eating disorders (EDs; n = 9 ARFID and n = 40 AN; R01MH103402) and ARFID (n = 42; R01MH108595). Participants had to be female and meet full criteria for ARFID or AN. Participant characteristics have been previously reported for a subset of participants (Aulinas et al., 2020; Becker et a., 2021; Breithaupt et al., 2020; Bryant-Waugh et al., 2019; Harshman et al., 2021; Harshman et al., 2019; Izquierdo et al., 2019; Kambanis et al., 2020; Kerem et al., 2021; Mancuso et al., 2020; Thomas et al., 2020; Wang et al., 2020) to test different hypotheses than explored in the current paper. Exclusion criteria for the parent studies included: (a) past-month active substance/ alcohol use disorders; (b) current/lifetime psychosis; and (c) intellectual disability (IQ < 70). Exclusion criteria pertaining to magnetic resonance imaging/neuroendocrine assessments are detailed elsewhere (Breithaupt et al., 2020; Mancuso et al., 2020); and (d) current active suicidality with plan/intent (current passive suicidality and history of active suicidality were allowed). Importantly, no participants had to be excluded due to current active suicidality with plan/intent.

Measures

Comorbid Psychiatric and ED Diagnoses—We used the Kiddie Schedule for Affective Disorder and Schizophrenia – Present and Lifetime (KSADS-PL) 2013 Working Draft (Kaufman et al., 2013) to assess history of depressive, anxiety, and neurodevelopmental disorders. We added questions keyed to *DSM-5* ARFID criteria.

Percent agreement between raters conferring diagnoses ranged from 94–100% (Kambanis et al., 2020; random sample representing 14% of R01MH108595).

We used the Pica, ARFID, and Rumination Disorder Interview (PARDI; Bryant-Waugh et al., 2019) to confer ARFID diagnoses in R01MH108595. We used the Eating Disorder Examination (EDE) and KSADS-PL to confirm ARFID diagnoses in R01MH103402. Interrater reliability for a subset of the ARFID sample was excellent (κ = 1.0; Thomas et al., 2020).

We used the EDE Version 17.0 (Fairburn et al., 2008) to confirm AN diagnosis in R01MH103402. Inter-rater reliability for a subset of the AN sample was excellent ($\kappa = 1.0$; Wang et al., 2020).

Suicidality—We used the Children's Depression Inventory 2 (CDI-2; Kovacs, 2011; a score of 2 ["I think about killing myself but I would not do it"] on Item 8) and the Beck Depression Inventory II (BDI-II; Beck et al., 1996; a score of 1 or 2 ["I have thoughts of killing myself, but I would not carry them out" or "I would like to kill myself"] on Item 9) to classify individuals as endorsing suicidality.

Procedure

Procedures were approved by the Mass General Brigham Institutional Review Board. Research coordinators administered the KSADS-PL at participants' screening visit. During participants' primary visit, dietitians measured participants' height/weight and study staff administered the PARDI/EDE.

Statistical Analyses

We used SPSS 28.0 (IBM Corp., 2021) for analyses. We used age at study presentation ("age") as our age variable due to its increased reliability (e.g., many individuals did not have an ARFID diagnosis preceding the study, some had significant symptoms prior to an official diagnosis). Prior to our main analyses, we conducted a t-test comparing age; females with ARFID were significantly younger (p < .001, Cohen's d = .94; Table 1), supporting our inclusion of age as a covariate.

Aim 1 – Omitting Age as a Covariate—To test our hypotheses that females with ARFID would demonstrate: (a) lower frequency of depressive disorders and suicidality; (b) higher frequency of anxiety disorders; and (c) higher frequency of neurodevelopmental disorders, we conducted chi square tests of independence for broad lifetime KSADS-PL categories and suicidality.

Aim 2 – Including Age as a Covariate—For exploratory analyses that included age as a covariate, we performed logistic regression. We performed analyses when expected cell counts were > 1 with less than 20% of cells having expected frequencies < 5 (Tabachnick & Fidell, 2013; Table 2). We simultaneously entered diagnosis and age as covariates and absence/presence of each relevant diagnosis as the binary criterion variable. We were unable to perform logistic regression for lifetime neurodevelopmental disorders because the cell count assumption was not met.

Results

Demographics and clinical characteristics are presented in Table 1.

Omitting Age as a Covariate

Frequencies of lifetime psychiatric comorbidities are presented in Table 2. Consistent with our hypotheses, results from chi square tests of independence indicated that lifetime depressive disorders ($\chi^2[1] = 4.99$, p = .025, Cramer's V = .23) were significantly less frequent among females with ARFID than AN. The chi square cell count assumption for suicidality was not met; a Fisher's exact test demonstrated that suicidality was significantly less frequent among females with ARFID than AN ($\chi^2[1] = 8.89$, p = .003, Cramer's V = .31). Contrary to our hypothesis, there were no between-group differences in anxiety disorders ($\chi^2[1] = 1.58$, p = .208, Cramer's V = .13). Finally, neurodevelopmental disorders were four times as frequent in ARFID group, though this trend was non-significant ($\chi^2[1] = 3.42$, p = .064, Cramer's V = .19).

Including Age as a Covariate

Results from logistic regression analyses are presented in Table 2. When including age as a covariate, no between-group differences emerged in depressive (60% AN v. 37% ARFID; p = .276) or anxiety (65% AN v. 53% ARFID; p = .938) disorders. Females with ARFID still had a lower frequency of suicidality than AN (35% AN v. 10% ARFID; p = .009).

Discussion

Given different ages of onset of specific psychiatric disorders and that ARFID typically develops in childhood whereas AN typically develops in adolescence or young adulthood, we compared the frequency of lifetime psychiatric comorbidities and suicidality between females with ARFID and AN omitting (Aim 1) and including (Aim 2) age as a covariate to improve our understanding of the impact of age on differential comorbidity profiles. Omitting age as a covariate, females with ARFID had a lower frequency of depressive disorders and suicidality, which was consistent with our hypothesis. Only the result regarding suicidality remained significant when including age as a covariate. These data suggest that suicidality is at least three times less common in ARFID than AN irrespective of age. However, observed differences in other psychiatric comorbidities may reflect ARFID's relatively younger age at treatment presentation compared to AN.

Restrictive eating predicts suicidality (Forrest et al., 2016; Wang et al., 2018) and may increase risk of suicidal ideation even beyond risk associated with other eating behaviors (Wang et al., 2019). Restriction in AN functions to control shape/weight; however, research suggests that individuals with AN may also engage in restriction with intention to harm themselves (Fox et al., 2019). Suicidality in AN, then, may be tied to judging one's self-worth predominantly based on appearance. The absence of overvaluation of shape/weight among individuals with ARFID could account for their lower frequency of suicidality. Further, restriction in ARFID serves a different function than restriction AN; therefore, correlates of restriction may also differ. Since ours is the first study to compare suicidality

between ARFID and AN, future research should explore this finding across age ranges, given that both disorders can present across the lifespan.

No difference emerged in the frequency of overall anxiety disorders; these are common psychiatric comorbidities for both EDs. There was a non-significant trend for neurodevelopmental disorders to be more frequent in ARFID; however, we were unable to examine the extent to which age may have influenced this finding, warranting further research. Neurodevelopmental disorders are more common in males (Kessler et al., 2005), and past studies that reported a higher frequency in ARFID (i.e., Lieberman et al., 2019; Nicely et al., 2014) had higher proportions of males than females with ARFID than AN. Future studies should explore potential differences in both sexes. Additionally, future studies should examine illness duration as a covariate, which has implications for understanding the etiology of psychiatric comorbidities in ARFID and AN as well as the temporal relationship between onset and the development of comorbidities.

Limitations of this study include that exclusion criteria for the parent study prevented us from comparing certain comorbidities. Though no participants were excluded due to current active suicidality, it is possible that individuals with current active suicidality chose not to participate in the study. Our sample was homogeneous (i.e., primarily non-Hispanic and White). Future research should focus on minoritized youth with AN and ARFID, given the dearth of research in this area, despite unique risk factors at play for these populations (Goel et al., 2020; Egbert et al., 2020, 2022). Our sample size was modest (N=91), making it challenging to ascertain whether our study was adequately powered to detect meaningful between-group differences. Some cell sizes were small, precluding logistic regression analyses for neurodevelopmental disorders. Finally, the inclusion of only females in our sample prevents the generalizability of our findings to males. Limitations notwithstanding, our findings suggest that previously reported differences in psychiatric comorbidities between ARFID and AN may be attributed, in part, to ARFID's younger age at clinical presentation. Suicidality appears to be the exception to this pattern of findings, being three times less common in ARFID than AN, regardless of age. Future research is needed to replicate these findings.

Financial support:

National Institute of Mental Health, Grant/Award Numbers: R01MH103402, R01MH108595, K23MH125143; K24MH120568.

Data availability statement:

All data will be made publicly available through the National Institute of Mental Health National Database for Autism Research at the conclusion of the study.

References

American Psychiatric Association. (2013). Diagnostic and statistical manual of mental disorders, 5th ed. (DSM-5). Arlington, VA: American Psychiatric Association.

Aulinas A, Marengi DA, Galbiati F, Asanza E, Slattery M, Mancuso CJ, Wons O, Micali N, Bern E, Eddy KT, Thomas JJ, Misra M, Lawson EA (2020). Medical comorbidities and endocrine

dysfunction in low-weight females with avoidant/restrictive food intake disorder compared to anorexia nervosa and healthy controls. International Journal of Eating Disorders, 53(4), 631–636. [PubMed: 32198943]

- Aulinas A, Plessow F, Pulumo RL, Asanza E, Mancuso CJ, Slattery M, Tolley C, Thomas JJ, Eddy KT, Miller KK, Klibanski A, Misra M, & Lawson EA (2019). Disrupted oxytocin-appetite signaling in females with anorexia nervosa. Journal of Clinical Endocrinology and Metabolism, 104(10), 4931–4940. [PubMed: 31251345]
- Beck AT, Steer RA, & Brown GK (1996). Manual for the Beck Depression Inventory-II. San Antonio, TX: Psychological Corporation.
- Becker KR, Mancuso C, Dreier MJ, Asanza E, Breithaupt L, Slattery M, Plessow F, Micali N, Thomas JJ, Eddy KT, Misra M, & Lawson EA (2021). Ghrelin and PYY in low-weight females with avoidant/restrictive food intake disorder compared to anorexia nervosa and healthy controls. Psychoneuroendocrinology, 129, 105243. [PubMed: 34049199]
- Becker KR, Keshishian AC, Liebman RE, Coniglio KA, Wang SB, Franko DL, Eddy KT, & Thomas JJ (2019). Impact of expanded diagnostic criteria for avoidant/restrictive food intake disorder on clinical comparisons with anorexia nervosa. International Journal of Eating Disorders, 52(3), 230–238. [PubMed: 30578644]
- Berkson J (1946). Limitations of the application of fourfold table analysis to hospital data. Biometrics Bulletin, 2(3), 47–53. [PubMed: 21001024]
- Breithaupt L, Chunga-Iturry N, Lyall AE, Cetin-Karayumak S, Becker KR, Thomas JJ, Slattery M, Makris N, Plessow F, Pasternak O, Holsen LM, Kubicki M, Misra M, Lawson EA, & Eddy KT (2020). Developmental stage-dependent relationships between ghrelin levels and hippocampal white matter connections in low-weight anorexia nervosa and atypical anorexia nervosa. Psychoneuroendocrinology, 119: 104722. [PubMed: 32512249]
- Bryant-Waugh R, Micali N, Cooke L, Lawson EA, Eddy KT, & Thomas JJ (2019). Development of the Pica, ARFID, and Rumination Disorder Interview, a multi-informant, semi-structured interview of feeding disorders across the lifespan: A pilot study for ages 10–22. International Journal of Eating Disorders, 52(4), 378–387. [PubMed: 30312485]
- Bryson AE, Scipioni AM, Essayli JH, Mahoney JR, & Ornstein MD (2018). Outcomes of low-weight patients with avoidant/restrictive food intake disorder and anorexia nervosa at long-term follow-up after treatment in a partial hospitalization program for eating disorders. International Journal of Eating Disorders, 51(5), 470–474. [PubMed: 29493804]
- Egbert AH, Haedt-Matt A, Smith KE, Culbert K, Engel S, & Goldschmidt AB (2020). Momentary associations between positive affect dimensions and dysregulated eating during puberty in a diverse sample of youth with overweight/obesity. International Journal of Eating Disorders, 53(10), 1667–1677. [PubMed: 32706499]
- Egbert AH, Hunt RA, Williams KL, Burke NL, & Jennings KM (2022). Reporting racial and ethnic diversity in eating disorder research over the past 20 years. International Journal of Eating Disorders, 1–8.
- Fairburn CG, Cooper Z & O'Connor M (2008). Eating Disorder Examination (Edition 17.0). In: Fairburn CG. Cognitive Behavior Therapy and Eating Disorders (pp. 265–308. New York, NY: Guilford Press
- Fisher MM, Rosen DS, Ornstein RM, Mammel KA, Katzman DK, Rome ES, Callahan ST, Malizio J, Kearney S, & Walsh BT (2014). Characteristics of avoidant/restrictive food intake disorder in children and adolescents: a "new disorder" in DSM-5. Journal of Adolescent Health, 55(1), 49–52.
- Forrest LN, Bodell LP, Witte TK, Goodwin N, Bartlett ML, Siegfried N, Eddy KT, Thomas JJ, Franko DL, & Smith AR (2016). Associations between eating disorder symptoms and suicidal ideation through thwarted belongingness and perceived burdensomeness among eating disorder patients. Journal of Affective Disorders, 195, 127–135. [PubMed: 26895090]
- Fox KR, Wang SB, Boccagno C, Haynos AF, Kleiman E, & Hooley JM (2019). Comparing self-harming intentions underlying eating disordered behaviors and NSSI: Evidence that distinctions are less clear than assumed. International Journal of Eating Disorders, 52(5), 564–575. [PubMed: 30770581]

Goel NJ, Burnette CB, Mazzeo SE (2019). Racial and ethnic differences in the association between parent-oriented perfectionism and disordered eating in college women. International Journal of Eating Disorders, 53(2), 191–200. [PubMed: 31593340]

- Harshman SG, Jo J, Kuhnle M, Hauser K, Murray HB, Becker KR, Misra M, Eddy KT, Micali N, Lawson EA*, & Thomas JJ* (2021). A moving target: How we define avoidant/restrictive food intake disorder can double its prevalence. Journal of Clinical Psychiatry, 82(5), 20m13831.*Cosenior authors.
- Harshman SG, Wons O, Rogers MS, Izquierdo AM, Holmes TM, Pulumo RL, Asanza E, Eddy KT, Misra M, Micali N, Lawson EA, & Thomas JJ (2019). A diet high in processed foods, total carbohydrates and added sugars, and low in vegetables and protein is characteristic of youth with avoidant/restrictive food intake disorder. Nutrients, 11(9), 2013. [PubMed: 31461865]
- IBM Corp. (2020). IBM SPSS statistics: Version 27. Armonk, NY: IBM Corp.
- Insel T, Cuthbert B, Garvey M, Heinssen R, Pine DS, Quinn K, Sanislow C, & Wang P (2010). Research domain criteria (RDoC): Toward a new classification framework for research on mental disorders. American Journal of Psychiatry, 167(7), 748–751. [PubMed: 20595427]
- Izquierdo A, Plessow F, Becker KR, Mancuso CJ, Slattery M, Murray HB, Hartmann AS, Misra M, Lawson EA, Eddy KT, & Thomas JJ (2019). Implicit attitudes toward dieting and thinness distinguish fat-phobic and non-fat-phobic anorexia nervosa from avoidant/restrictive food intake disorder in adolescents. International Journal of Eating Disorders, 52(4), 419–427. [PubMed: 30597579]
- Joiner T (2005). Why people die by suicide. Harvard University Press.
- Kambanis PE, Kuhnle MC, Wons OB, Jo JH, Keshishian AC, Hauser K, Becker KR, Franko DL, Misra M, Micali N, Lawson EA, Eddy KT, & Thomas JJ (2020). Prevalence and correlates of psychiatric comorbidities in children and adolescents with full and subthreshold avoidant/restrictive food intake disorder. International Journal of Eating Disorders, 52(2), 256–265.
- Kaufman J, Birmaher B, Axelson D, Perepletchikova F, Brent D, & Ryan N (2013). Kiddie Schedule for Affective Disorders and Schizophrenia Present and Lifetime Version 2013: Working Draft (K-SADS-PL).
- Keery H, LeMay-Russell S, Barnes TL, Eckhardt S, Peterson CB, Lesser J, Gorrell S, & Le Grange D (2019). Attributes of children and adolescents with avoidant/restrictive food intake disorder. Journal of Eating Disorders, 7, 31. [PubMed: 31528341]
- Kerem L, Van de Water AL, Kuhnle MC, Harshman S, Hauser K, Eddy KT, Becker KR, Misra M, Micali N, Thomas JJ, Holsen L, & Lawson EA (2021). Neurobiology of avoidant/restrictive food intake disorder in youth with overweight/obesity versus healthy weight. Journal of Clinical Child and Adolescent Psychology, 53, 1–14.
- Kessler RC, Berglund P, Demler O, Jin R, Merikangas KR & Walters EE (2005). Lifetime prevalence and age-of-onset distributions of DSM-IV disorders in the National Comorbidity Survey Replication. Archives of General Psychiatry, 62(6), 593–602. [PubMed: 15939837]
- Kotov R, Krueger RF, Watson D, Achenbach TM, Althoff RR, Bagby RM, Brown TA, Carpenter WT, Caspi A, Clark LA, Eaton NR, Forbes MK, Forbush KT, Goldberg D, Hasin D, Hyman SE, Ivanova MY, Lyman DR, Markon K, Miller JD, ... & Zimmerman M (2017). The Hierarchical Taxonomy of Psychopathology (HiTOP): A dimensional alternative to traditional nosologies. Journal of Abnormal Psychology, 126(4), 454–477. [PubMed: 28333488]
- Kovacs M (2001). Children's depression inventory manual. Multi health systems North Tonawanda: Multi Health Systems, NY.
- Kraemer HC (2015). A source of false findings in published research studies: Adjusting for covariates. JAMA Psychiatry, 72(10), 961–962. [PubMed: 26244634]
- Lawrence HR, Burke TA, Sheehan AE, Pastro B, Levin RY, Walsh RFL, Bettis AH, & Liu RT (2021).
 Prevalence and correlates of suicidal ideation and suicide attempts in preadolescent children: A US population-based study. Translational Psychiatry, 11.
- Lieberman M, Houser ME, Voyer A, Grady S, & Katzman DK (2019). Children with avoidant/ restrictive food intake disorder and anorexia nervosa in a tertiary care pediatric eating disorder program: A comparative study. International Journal of Eating Disorders, 52, 239–245. [PubMed: 30706952]

Mancuso C, Izquierdo A, Slattery M, Becker K, Plessow F, Thomas JJ, Eddy K, Lawson EA, & Misra M (2020). Changes in appetite-regulating hormones following food intake are associated with changes in reported appetite and a measure of hedonic eating in girls and young women with anorexia nervosa. Pscyhoneuroendocrinology, 113:104556.

- Nicely TA, Lane-Loney S, Masciulli E, Hollenbeak CS, & Ornstein RM (2014). Prevalence and characteristics of avoidant/restrictive food intake disorder in a cohort of young patients in day treatment for eating disorders. Journal of Eating Disorders, 2(21), 21. [PubMed: 25165558]
- Ornstein RM, Essayli JH, Nicely TA, Masciulli E, & Lane-Loney S (2017). Treatment of avoidant/restrictive food intake disorder in a cohort of young patients in a partial hospitalization program for eating disorders. International Journal of Eating Disorders, 50(9), 1067–1074. [PubMed: 28644568]
- Solmi M, Radua J, Olivola M, Croce E, Soardo L, Salazar de Pablo G, Il Shin J, Kirkbride JB, Jones P, Kim JH, Kim JY, Carvalho AF, Seeman MV, Correll CU, & Fusar-Poli P (2022). Age at onset of mental disorders worldwide: Large-scale meta-analysis of 192 epidemiological studies. Molecular Psychiatry, 27(1), 281–295. [PubMed: 34079068]
- Striegel-Moore RH, & Bulik CM (2007). Risk factors for eating disorders. American Psychologist, 62, 181–198. [PubMed: 17469897]
- Tabachnick B, & Fidell L (2013). Using Multivariate Statistics (6th ed). Boston: Pearson.
- Thomas JJ, Becker KR, Kuhnle MC, Jo JH, Harshman SG, Wons OB, Keshishian AC, Hauser K, Breithaupt L, Liebman RE, Misra M, Wilhelm S, Lawson EA, & Eddy KT (2020).
 Cognitive-behavioral therapy for avoidant/restrictive food intake disorder (CBT-AR): Feasibility, acceptability, and proof-of-concept for children and adolescents. International Journal of Eating Disorders, 53(10), 1636–1646. [PubMed: 32776570]
- Thomas J, & Eddy K (2019). Cognitive-Behavioral Therapy for Avoidant/Restrictive Food Intake Disorder: Children, Adolescents, and Adults. Cambridge, UK: Cambridge University Press.
- Wang SB, Mancuso CJ, Jo J, Keshishian AC, Becker KR, Plessow F, Izquierdo AM, Slattery M, Franko DL, Misra M, Lawson EA, Thomas JJ, & Eddy KT (2020). Restrictive eating, but not binge eating or purging, predicts suicidal ideation in adolescents and young adults with low-weight eating disorders. International Journal of Eating Disorders, 53(3), 472–477. [PubMed: 31886575]
- Wang SB, Pisetsky EM, Skutch JM, Fruzzetti AE, & Haynos AF (2018). Restrictive eating and nonsuicidal self-injury in a nonclinical sample: Co-occurrence and associations with emotion dysregulationa dn interpersonal problems. Comprehensive Psychiatry, 82, 128–132. [PubMed: 29477705]
- Zanna V, Criscuolo M, Mereu A, Cinelli G, Marchetto C, Pasqualetti P, Tozzi AE, Castiglioni MC, Chianello I, & Vicari S (2020). Restrictive eating disorders in children and adolescents: A comparison between clinical and psychopathological profiles. Eating and Weight Disorders, 26(5), 1491–1501. [PubMed: 32720247]

Table 1

Demographics and clinical characteristics of 51 females with avoidant/restrictive food intake disorder and 40 females with anorexia nervosa

Demographic and clinical characteristics	ARFID	AN
	M (SD)	M (SD)
Age (years) ¹	16.61 (3.92)	19.59 (2.16)
	n, (%)	n, (%)
Race		
Black/African American	1 (2)	0 (0)
Asian	3 (6)	7 (18)
White	44 (85)	32 (80)
Other	4 (8)	1 (2)
Ethnicity		
Hispanic	7 (13)	4 (10)
Non-Hispanic	45 (87)	36 (90)
ARFID Profile ²		
Sensory sensitivity	36 (69)	N/A
Fear of aversive consequences	13 (25)	N/A
Lack of interest in food/eating	29 (56)	N/A

 $\textit{Note}. \ ARFID-avoidant/restrictive \ food \ intake \ disorder; \ AN-anorexia \ nervosa; \ M-mean; \ SD-standard \ deviation; \ N/A-not \ applicable.$

 $^{^{}I}$ There was a statistically significant difference in age between ARFID and AN (p < .001, Cohen's d = .94).

 $^{^2}$ We used the PARDI to determine presence/absence of each ARFID profile for n = 42 individuals in study R01MH108595. The n = 9 individuals with ARFID who we recruited as part of study R01MH103402 are not represented in this table because we did not administer the PARDI to participants in this study. Combined percentages exceed 100% because, consistent with Thomas and colleagues' (2017) three-dimensional model of ARFID, participants could present with more than one profile.

Author Manuscript

Author Manuscript

Kambanis et al. Page 11

Table 2

Frequency and results of logistic regression analyses, including age as a covariate, for lifetime psychiatric comorbidities and suicidal ideation in 51 females with avoidant/restrictive food intake disorder and 40 females with anorexia nervosa.

Lifetime KSADS-PL Diagnoses (Binary Criterion Variables) I	ARFID n (%)	AN n (%)	В	SE	Wald χ^2	Wald \mathcal{X}^2 p-value	OR [95% CI]
Depressive and Bipolar-Related Disorders ***	19 (37)	24 (60)	52	.48	1.19	.276	.59 [.23, 1.52]
Major depressive disorder	15 (29)	24 (60)					
Other specified depressive disorder	3 (6)	0 (0)					
Persistent depressive disorder	1 (2)	0 (0)					
Anxiety, Obsessive-Compulsive, and Trauma-Related Disorders	27 (53)	26 (65)	90.	.51	.01	.938	1.04 [.39, 2.80]
Generalized anxiety disorder	18 (35)	18 (45)					
Panic disorder	7 (14)	3 (8)					
Specific phobia	5 (10)	5 (13)					
Social anxiety disorder	5 (10)	15 (38)					
Separation anxiety disorder	0 (0)	4 (10)					
Other specified anxiety disorder	1 (2)	0 (0)					
Obsessive compulsive disorder	3 (6)	2 (5)					
Posttraumatic stress disorder	4 (8)	2 (5)					
Neurodevelopmental, Disruptive, and Conduct Disorders 2	7 (14)	1 (3)					
Oppositional defiant disorder	1 (2)	0)0					
Autism spectrum disorder	1 (2)	0 (0)					
Attention deficit/hyperactivity disorder	4 (8)	1 (3)					
Other specified attention deficit/hyperactivity disorder	2 (4)	0 (0)					
Suicidality 3***	5 (10)	14 (35)	-1.75	.67	6.87	* 600°	.18 [.05, .64]

Note. ARFID – avoidant/restrictive food intake disorder; AN – anorexia nervosa; KSADS-PL – Kiddie Schedule for Affective Disorders and Schizophrenia – Present and Lifetime Version; SE – standard error; OR - odds ratio. Anorexia nervosa is the reference group.

frequencies <5 [Tabachnick & Fidell, 2013]) or those under broad KSADS-PL categories that were not significant (i.e., Anxiety, Obsessive-Compulsive, and Trauma-Related Disorders) and were therefore Individual diagnoses with no test statistics reported indicate those that did not have sufficient representation in all cells for logistic regression (i.e., > 1 with less than 20% of cells having expected not analyzed.

² Assumptions for logistic regression analysis were not met and therefore, this test was not performed on this category of disorders.

 $[\]mathfrak{F}$ The self-report measures we used to assess suicidality reflect the past two weeks

** Represents KSADS-PL categories that were statistically significant in chi-square tests of independence that omitted age as a covariate.

* Bolded p-values indicate those that met the threshold for statistical significance of p < .05.

Page 12